Author's response to reviews

Title: High number of CD45RO+ tumor infiltrating lymphocytes is an independent prognostic factor in non-metastasized (stage I-IIA) esophageal adenocarcinoma

Authors:

Sandra Rauser (sandra.rauser@helmholtz-muenchen.de)
Rupert Langer (Rupert.Langer@lrz.tum.de)
Sebastian Tschernitz (sebastian.tschernitz@mytum.de)
Peter Gais (peter.gais@mnet-mail.de)
Uta Jütting (uta.juetting@helmholtz-muenchen.de)
Marcus Feith (feith@chir.med.tu-muenchen.de)
Heinz Höfler (Hoefler@lrz.tu-muenchen.de)
Axel Walch (axel.walch@helmholtz-muenchen.de)

Version: 4 Date: 15 October 2010

Author's response to reviews:

Dear editor,

enclosed you will find the final version of our manuscript. We performed all requested formatting changes, in detail:

1 - shading within the tables was removed
2 - abstract: keywords were removed
3 - tables: vertical lines were removed
4 - figures: figure numbers within the figures were removed
5 - last check for typographic errors was performed

kind regards,

R. Langer and A. Walch